\documentclass[11pt]{amsart}
\usepackage{amssymb,times}
\begin{document}

\title{PROBABILISTIC ASPECTS OF FINITE SIMPLE GROUPS, AND APPLICATIONS}
\author{Matrin Liebeck}

\maketitle

\pagenumbering{}

The course will cover various probabilistic and asymptotic aspects
of finite groups. The focus will be on simple groups, but there
will be applications to general finite groups, permutation groups,
and further afield to Fuchsian groups, representation varieties,
word maps on groups, and growth in Cayley graphs.

\end{document}
